Author's response to reviews

Title: Methodological aspects of a GIS-based environmental health inspection program used in the Athens 2004 Olympic and Para Olympic Games

Authors:

Christos Hadjichristodoulou (hadjich@ath.forthnet.gr)
Elpidoforos S Soteriades (esoteria@hsph.harvard.edu)
Virginia Kolonia (hadjich@ath.forthnet.gr)
Matthew E Falagas (matthew.falagas@tufts.edu)
Efstathios Pantelopoulos (hadjich@ath.forthnet.gr)
Georgios Panagakos (hadjich@ath.forthnet.gr)
Varvara Mouchtouri (hadjich@ath.forthnet.gr)
Jeni Kremastinou (hadjich@ath.forthnet.gr)

Version: 6 Date: 21 August 2005

Author's response to reviews:

Dear editor,

We would like to change the affiliation of the co-author Matthew Falagas. He will have the same affiliation with the other authors National School of Public Health. Thus in the manuscript will appear only two affiliations the University of Thessaly and the National School of Public Health.